\begin{thebibliography}{10}

\bibitem{marshall}
H.~Jones and J.~Marshall, ``{Convection with rotation in a neutral ocean: a
  study of open deep convection},'' {\em {Journal of Physical Oceanography}},
  vol.~23, no.~6, pp.~1009--1039, 1993.

\bibitem{GCM}
Arvind, K.-C. Cho, C.~Hill, R.~P. Johnson, J.~Marshall, and A.~Shaw, ``{A
  Comparison of Implicitly Parallel Multithreaded and Data Parallel
  Implementations of an Ocean Model based on the Navier-Stokes Equations},''
  Tech. Rep. CSG Memo 364, Laboratory for Computer Science, MIT, 1991.
\newblock {\small http://www-csg.lcs.mit.edu/GCM/}.

\bibitem{id}
R.~S. Nikhil, ``{Id Language Reference Manual, Version 90.1},'' Tech. Rep. CSG
  Memo 284-2, Laboratory for Computer Science, MIT, July 1991.

\bibitem{monsoon}
G.~M. Papadopoulos, {\em {Implementation of a General-Purpose Dataflow
  Multiprocessor}}.
\newblock Research Monograph in Parallel and Distributed Computing, MIT Press,
  1992.

\bibitem{frame-manager}
D.~Chiou, ``{Activation Frame Memory Management for the Monsoon Processor},''
  Master's thesis, MIT EECS, September 1992.
\newblock {\small http://www-csg.lcs.mit.edu/$\sim$derek/MS-Thesis.ps}.

\bibitem{Culler90}
D.~E. Culler, {\em {Managing Parallelism and Resources in Scientific Dataflow
  Programs}}.
\newblock PhD thesis, MIT EECS, 545 Technology Square, Cambridge, MA 02139,
  1990.
\newblock LCS TR-446.

\bibitem{Wolfe95}
M.~Wolfe, {\em High Performance Compiler for Parallel Computing}.
\newblock Addison Wesley, 1995.

\bibitem{boon-loops}
B.~S. Ang, ``Efficient implementation of sequential loops in dataflow
  computation,'' in {\em Proceedings of the Functional Programming Computer
  Architecture}, June 1993.
\newblock {\small ftp://csg-ftp.lcs.mit.edu/pub/papers/misc/ seq-loop.ps.Z}.

\bibitem{templates}
R.~Barrett, M.~Berry, T.~F. Chan, J.~Demmel, J.~Donato, J.~Dongarra,
  V.~Eijkhout, R.~Pozo, C.~Romine, and H.~Van~der Vorst, {\em Templates for the
  Solution of Linear Systems: Building Blocks for Iterative Methods}.
\newblock Philadelphia, PA: SIAM, 1994.
\newblock {\small http://www.netlib.org/linalg/html\_templates/
  Templates.html}.

\bibitem{CMF}
Thinking Machine Corporation, {\em CM Fortran Language Reference Manual, V2.1},
  January 1994.

\bibitem{CM-5}
W.~D. Hillis and L.~W. Tucker, ``The {CM-5} connection machine: {A} scalable
  supercomputer,'' {\em Communications of the ACM}, vol.~36, pp.~31--40, Nov.
  1993.

\bibitem{shaw92b}
A.~Shaw, Y.~Kodama, M.~Sato, S.~Sakai, and Y.~Yamaguchi, ``{Performance of
  Data-Parallel Primitives on the EM-4 Dataflow Parallel Supercomputer},'' in
  {\em Proceedings of Frontiers '92}, (McLean, VA), pp.~302--309, October 1992.

\bibitem{EM4a}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, Y.~Kodama, and T.~Yuba, ``{An Architecture
  of a Dataflow Single Chip Processor},'' in {\em Proceedings of ISCA},
  pp.~46--53, 1989.

\bibitem{Grafe88a}
V.~Grafe, G.~Davidson, J.~Hoch, and V.~Holmes, ``{The Epsilon Dataflow
  Processor},'' in {\em Proceedings of the 16th. Annual International Symposium
  on Computer Architecture, Jerusalem, Israel}, pp.~36--45, May 29-31 1989.

\bibitem{Mohr91}
E.~Mohr, D.~Kranz, and R.~Halstead~Jr., ``{Lazy task creation a technique for
  increasing the granularity of parallel programs},'' {\em IEEE Transactions on
  Parallel and Distributed Computing}, vol.~2, pp.~264--80, July 1991.
\newblock {\small ftp://cag.lcs.mit.edu/pub/papers/futures.ps.Z}.

\bibitem{Poly87}
C.~D. Polychronopoulos and D.~J. Kuck, ``{Guided self-scheduling: a practical
  scheduling scheme for parallel supercomputers},'' {\em {IEEE Transactions on
  Computers}}, vol.~C-36, pp.~1425--39, December 1987.

\bibitem{Arvind88b}
Arvind, D.~E. Culler, and K.~Ekanadham, ``The price of asynchronous
  parallelism: an analysis of dataflow architectures,'' in {\em Proceedings of
  CONPAR 88}, pp.~541--555, Sept. 1988.

\bibitem{id-performance}
J.~Hicks, D.~Chiou, B.~S. Ang, and Arvind, ``{Performance Studies of the
  Monsoon Dataflow Processor},'' {\em Journal of Parallel and Distributed
  Computing}, vol.~18, pp.~273--300, July 1993.
\newblock {\small
  http://www-csg.lcs.mit.edu/$\sim$derek/Monsoon-Performance.ps}.

\bibitem{Yeung93}
D.~Yeung and A.~Agarwal, ``{Experience with Fine-Grain Synchronization in MIMD
  Machines for Preconditioned Conjugate Gradient},'' in {\em {Proceedings of
  the Fourth Symposium on Principles and Practices of Parallel Programming}},
  pp.~{187--197}, {May} 1993.
\newblock {\small ftp://cag.lcs.mit.edu/pub/papers/fine-grain.ps.Z}.

\bibitem{Sur94}
S.~Sur and W.~Bohm, ``{Functional, I-structure, and M-structure implementations
  of NAS benchmark FT},'' in {\em Parallel Architectures and Compilation
  Techniques}, pp.~47--56, August 1994.
\newblock {\small http://www.colostate.edu/ $\sim$dataflow/papers/pact94b.ps}.

\end{thebibliography}

\Section{Introduction}

Dataflow machines have two basic types of overheads which do not exist in
sequential machines: {\it synchronization} and {\it control} overheads.
Synchronization overheads are primarily encountered within basic blocks,
and control overheads are encountered between basic blocks (on conditional,
loop, and procedure call boundaries).  These overheads are the cost of
exploiting fine-grained parallelism.  In this paper we consider tuning
techniques to alleviate these overheads for regular, {\it scientific}
applications on a dataflow computer.  These tuning techniques apply to both
compiler-directed optimizations as well as programmer-directed
optimizations.

\begin{fig}{figure:ocean}{In an ocean modeling code, the ocean is
represented by multiple three-dimensional arrays holding values for
different physical characteristics of the ocean.  Typically, a large
percentage of the geometry represents land, which is not of interest.}
\illustrationx{3.2in}{figures/ocean.eps}
\end{fig}

Many scientific applications spend much of their time in loops traversing
over large arrays which represent physical characteristics distributed over
the geometry of the problem.  One such application is the MIT ocean
circulation model GCM \cite{marshall} \cite{GCM} whose data parallel
implementation on the CM-5 has shown very good performance.  While
developing GCM for the CM-5, we also wrote a version of it in Id \cite{id}
for the Monsoon dataflow machine \cite{monsoon} \cite{CulPap90}.  Id is an
implicitly parallel language with a multithreaded implementation model, and
the GCM code allowed us to compare the execution overheads for the
multithreaded and data parallel models on an application which was
extremely well-suited for data parallel execution.  A more in-depth study
of the data parallel and multithreaded implementations of GCM is presented
in \cite{GCM}, where we compare program styles, ease of programming, and
algorithmic issues, while in this paper, we primarily discuss performance
and tuning on the dataflow platform.

Figure \ref{figure:ocean} shows a typical problem geometry -- the Pacific
ocean basin -- used in the circulation model.  A Pacific ocean state array
would typically contain 168 $\times$ 90 $\times$ 10 elements, 40\% of which
represent land, and do not contribute to the simulation.  However, unlike
some physical models, geometries used in ocean modeling are usually not
irregular enough to warrant repartitioning to eliminate land elements.

The core of GCM code is a preconditioned conjugate gradient (PCG) solver
which takes 80\% or more of the computational time of the application.  The
PCG consists of simple operations over entire state arrays which can be
described as triply-nested loops with relatively small loop bodies.  In
order to attain high performance for this application, we must execute
these loops efficiently while also distributing the work among the
processors.  Dataflow architectures differ significantly from conventional
sequential microprocessors or vector processors, so the process of tuning
these loops to obtain high performance is also quite different than tuning
loops written in conventional languages for conventional architectures.

\SubSection{Id: an implicitly parallel language}

Id is an implicitly parallel language which can express parallelism at the
instruction-, procedure-, and loop-level.  Id programs are compiled into
{\it dataflow graphs}, which are optimized, and then executed directly by
the Monsoon architecture.

\paragraph*{Instruction-level parallelism.}

\begin{fig}{figure:dataflow}{Optimized Monsoon dataflow graph for the body 
of the innermost loop of a three-dimensional inner product code.  It
directly represents instruction-level parallelism, but incurs fine-grained
synchronization costs.}
\illustrationx{2.9in}{figures/dfg-3d.ps}
\end{fig}

Dataflow graphs such as in Figure \ref{figure:dataflow} directly represent
instruction-level parallelism -- in the Monsoon architecture, such
instruction-level parallelism is only exploited within a processor.  Basic
blocks are represented by acyclic graphs such as in Figure
\ref{figure:dataflow}.  The {\it fork} operator in the figure is necessary
for sending values to more than two instruction at a time -- because of
opcode encoding limitations, most Monsoon instructions can send a value to
two destination instructions, and some instructions can send a value to
only one instruction.

Memory operations such as {\it i-fetch} are split-phase -- the initiation
and response are separate, and the latency of the response may be very
long.  Each memory operation has an implicit synchronization associated
with it so that reads which are executed before the write to the same
memory location wait for the write to occur before responding.

Compile-time constants (i.e.\ literals) and run-time constants (such as
loop constants) are represented in Figure \ref{figure:dataflow} in light
gray.  Run-time constants must be stored into local memory at run-time
using the {\it constant-store} instruction, and literals are stored into
local memory at load-time by the loader.  Both types of constants can be
accessed without synchronization overhead.

Control is implemented by using the {\it switch} operator which steers an
input value to one of two locations depending upon the value of an input
predicate.  Switch operators can be used to build conditionals or loops,
but incur high overhead because a different switch operator is needed for
each loop or conditional variable.

\paragraph*{\mbox{Procedure-level parallelism.}}

\hfill\mbox{Procedure-level} parallelism is expressed by implicitly forking
every procedure call and joining on the return value of the procedure.
Each procedure requires an {\it activation frame}, which is used to store
intermediate results and as a synchronization namespace for instructions in
the procedure.  An activation frame is roughly analagous to a stack frame
in a sequential processor.

Each procedure call requires a frame allocation and deallocation.  A frame
may be allocated on the local processor or on a remote processor to expoit
inter-processor parallelism, and frame allocation and deallocation is
handled by the run-time system.  Care has been taken to eliminate run-time
system costs, to the point where much of the run-time system is written in
assembly, and supported by special ``microcoded'' machine instructions
\cite{frame-manager}.  In the default run-time system, frame memory on each
processor is divided equally among all processors, so that each processor
manages part of the frame space of every other processor.  Frame allocation
requests are then handled locally in a round-robin fashion across all of
the processors in an attempt to provide some form of load balancing.

Despite the care taken in designing the run-time system, frame management
in this parallel environment is still more expensive than in a sequential
environment, where a simple pointer-bump can allocate or deallocate a stack
frame.  Furthermore, every argument is sent as a separate message, and
although the composition, sending and receiving of these messages are well
supported in hardware, argument-passing is still relatively more expensive
than register-style argument passing in sequential processors.  In general,
the programmer should inline most non-recursive procedures to avoid these
twin overheads of procedure calls.

\paragraph*{Loop-level parallelism.}

Loop-level parallelism can be exploited in several ways.  A direct
tail-recursive implementation of loops will use too many frame resources,
so most loops are either ``sequential'' loops or $k$-bounded loops
\cite{Culler90}.  Sequential loops execute on one processor, and only
exploit instruction-level parallelism; $k$-bounded loops execute $k$
consecutive iterations worth of parallelism either within or across
processors.

\begin{fig}{figure:dataflow-loop}{Schema for ``sequential'' dataflow loops,
which incur high fine-grained synchronization costs because of the need for
switch and fork operators.}
\illustrationx{2.9in}{figures/dataflow-loop.eps}
\end{fig}

Figure \ref{figure:dataflow-loop} shows the schema for a sequential loop
for Monsoon.  A number of input variables and loop constants are initially
fed to the loop body.  Their values for the next iteration are either fed
back into the top of the loop body, or else output to the final values.
The values are steered to their appropriate destination through the {\it
switch} dataflow instruction.  Since the switch instruction can only steer
one value to one destination at a time, an additional fork may be required
to distribute the loop test predicate value for each additional loop
variable.  Loop constants are stored into the frame before loop execution
and extracted after loop completion.

The $k$-bounded loop schema is more complicated than the sequential schema,
and employs a ring of $k$ activation frames which are allocated and linked
at the beginning of the loop and deallocated at the end of the loop.  Loop
constants must be stored into and extracted from each frame in a
$k$-bounded loop.  Induction variables are passed from iteration to
iteration like arguments are passed for procedure calls.

{\it Standard loop optimizations such as unrolling, peeling, strip-mining,
and interchange can have a larger impact on performance than for
conventional architectures because of the overhead of parallel asynchronous
execution of loop iterations in the dataflow model.}

High loop overhead for dataflow execution is exacerbated in Id because
array extents in Id, unlike Fortran, can be computed at run-time.
Consequently, to perform many loop optimizations important for performance
on scientific codes, one would need pragmas in Id to indicate static array
extents.  However, this type of overhead is due to the language, and not
dataflow execution.

\SubSection{The Monsoon dataflow machine}

\begin{fig}{figure:monsoon}{Monsoon consists of processing elements (PE's)
connected to I-structure memory modules (IS's) through a multi-stage
network.  Each PE is an eight-stage pipelined dataflow processor.}
\illustrationx{3.2in}{figures/monsoon.eps}
\end{fig}

Figure \ref{figure:monsoon} (taken from \cite{id-performance}) shows a
high-level view of the Monsoon multiprocessor.  Monsoon \cite{monsoon}
consists of eight 64-bit processing elements (PE's) and eight I-structure
memory modules (or IS's) connected via a multistage packet switch network.
Each PE is an eight-stage pipelined processor.  On each processor cycle, a
token enters the pipeline and its result is available eight cycles later.
Keeping an eight processor Monsoon busy requires at least 64-fold
parallelism because each stage of the pipeline executes a different thread.
Each PE can process ten million tokens per second and has 256K 64-bit words
of local memory on which frames are allocated.  In the memory module where
heap objects reside, word-level presence bits are implemented in hardware
to support fine-grain synchronizing data structures for global heap.  Each
IS has 4M 64-bit words of heap memory.  Monsoon's network interface with
bandwidth of 100M bytes per second can deliver a token to the network every
cycle.

Each instruction executes in one or two cycles, including floating point,
integer, memory, control, and message instructions.  Any instruction which
has two non-constant inputs must synchronize on the arrival of its two
inputs, and requires two cycles to execute, one of which is counted as a
{\it bubble}.

Every global memory access goes over the network, and all locations in the
global memory are equidistant.  The mapping of work to processors is
governed by the frame allocation policy.  Heap objects are interleaved by
hardware across the IS's and in general, data mapping has no effect on
performance.  Monsoon accesses global memory in a split-phase manner, and
can do useful work during the long-latency of a memory access.  However,
this requires additional instruction-level parallelism to keep the
processor fully utilized.  Each Monsoon processor has 16 counters which can
keep accurate counts of various types of instructions executed.
